Author’s response to reviews

Title: Effectiveness of therapeutic footwear for children: A systematic review

Authors:
Matthew Jason Hill (matthew.hill@research.staffs.ac.uk;hill523@hotmail.com)
Aoife Healy (A.Healy@staffs.ac.uk)
Nachiappan Chockalingam (N.Chockalingam@staffs.ac.uk)

Version: 3 Date: 28 Apr 2020

Author’s response to reviews:

• Thank you for your consideration of our review please find our response to these below.

Reviewer reports:

Editorial Review.

Apologies for not responding to this in your prior correspondence but please review use of sex and gender (as previously flagged by your reviewer) in your manuscript. Sex refers to the biological and physiological characteristics that define men and women, whereas gender refers to the socially constructed roles, behaviours, activities, and attributes that a given society considers appropriate for men and women. Therefore, when describing your sample characteristics, please use sex rather than gender.

• The text has been modified in the tables and manuscript to address this comment.